Notes from the RICH 1 phone meeting 02/03/07

People Present:

Bristol: Richard.

CERN: Carmelo, Kenneth, Olav, and Richard.

Imperial: Bill and Dave.

Oxford: Tony.

Rutherford: Derek and Mike.

Apologies: Fabio, Gabriel, Matthew, Neville, Nick, Trevor and Werner.

1. Gas Enclosure.

The large leak has been identified without the need for large scale dismantling.  The source of the leak was identified as the witness holes outside the VELO exit window seal.  The two holes are now plugged and attempts have been made to measure the leak rate.  These have been hindered by varying atmospheric pressure, but an upper limit of 1 litre/hour has been set.

We conclude that the seal to the exit window to the beampipe has been made successfully.  There is still work to do on the GE mounts and restraints.  Currently counter weights are still mounted on the C-side door. They should remain until the C-side floor is replaced.

2. Quartz Window.

The window frames are not yet ready for manufacture.  (action Dave, Trevor to report on schedule). The bonding of the windows will be done following delivery of the frames (>May).  There is no requirement to have the windows to install the HPDs, but they are needed before installation of the MDMS.

3. Spherical Mirrors.

 The four mirrors have now been coated at SESO and will be retrieved and brought to CERN in the week beginning the 12th of March.  The D0 and radius of curvature measurements will be repeated once the mirrors are at CERN.  

4. Mirror Installation tooling and alignment jig.
The alignment rig and the tooling for the spherical mirrors have been completed in Oxford and are ready for shipping.  All that is needed is a quotation for the shipping from Luker Brothers, but it is expected that shipping will take place at the end of the week beginning 05/03/07.  Bridge pieces will be shipped with the rig, but no rails.  It is designed to use the same rails as the gas enclosure.  Copies of these will be needed.

Dave congratulated Matthew, Joe, Tony and all in the Oxford workshop who have completed this work to a very tight schedule.

The bottom rails for the spherical mirrors are now in the workshop at IC as are brackets and V-running mechanisms.  No jigging has been done to align the support lugs on the CMA frame and so the components will not form a complete plane. Geoff will visit CERN to measure up the lugs on a surface table then tailor-make the V-blocks.

Work is still being done on the lower flat mirror plates and some modification is needed (action IC). 3 adjusters are needed (action Richard).   A trial installation is desired.

Flat mirror installation boxes are in the hands of the manufacturers.  Nothing is known beyond the original six week estimation for the order.

The rails for the boxes are not finished.  Work is needed on mountings.  8 rails are needed in total.  1 set of bridging pieces is needed (action IC)

5. HPD assembly.

The quote from NAB has been received.  2 more companies have been asked for quotes, and these are expected by the 8th of March.  Orders can be made immediately after this.  The time scale for NAB is six weeks.

HPD assembly for the lower chariot is expected to be similar as for the upper chariot except that the position of the wheels needs to be changed.

(Action Bill, to prepare the method and safety documentation (PSSPS) for the installation of the flat mirrors, the spherical mirrors and the HPD modules).

6. Column dressing.

Four columns are completed.  A fifth is having the fibring and soldering done on the day of the meeting (02/03/07) and will be tested the week beginning 05/03/07.  More HPDs will be arriving in a couple of weeks, and more HV boards during LHCb week.  After this work will be put on hold until more HPDs arrive at the end of April/beginning of May.  The top half should be finished by the beginning of April.

All mechanics have been delivered apart from 2 end pieces that require anodizing. They will be at CERN in April.  Tests have been made on the cooling plates including leak testing on all of them using pressurised gas.  

7. Magnetic distortion monitoring system.

The following was discussed at the last meeting on the MDMS on the 22nd February:  There is one possible manufacturer, Parker, for the stepper motor and stage.  It is proposed that the stepper motor will go along the short side, and there will be a carbon-fibre bar along the long side on which will be mounted the light source.  This will be a flexible printed circuit board with LEDs on it.  Parker have given a preliminary estimate of ~ 5K CHF for each of the upper and lower stages.  The hardware required is non-standard in that the motor is rotated by 900 with respect to the gearbox.  There will be a further meeting on the 8th March.  In addition Ray will be invited to commissioning meetings.

8. Mirror monitoring.

All cameras have been purchased for RICH1.  The fibres have been installed.  The laser has been purchased.  It is the same laser as is used for RICH2.

